Author's response to reviews

Title: G betagamma subunits inhibit Epac-induced melanoma cell migration

Authors:

Erdene Baljinnym (baljiner@umdnj.edu)
Masanari Umemura (umemurma@umdnj.edu)
Mariana S DeLorenzo (delorems@umdnj.edu)
Lai-Hua Xie (xiela@umdnj.edu)
Martha Nowycky (Martha.Nowycky@umdnj.edu)
Mizuka Iwatsubo (mintz999@gmail.com)
Susie Chen (suziec@rci.rutgers.edu)
James S Goydos (goydosjs@umdnj.edu)
Kousaku Iwatsubo (iwatsuko@umdnj.edu)

Version: 8 Date: 14 June 2011

Author’s response to reviews: see over
Dear Editor,

I am very glad that our manuscript entitled “Gβγ subunits inhibit Epac-induced melanoma cell migration” is accepted for publication in BMC Cancer.

We are re-submitting the manuscript proofread according to the suggestions made by the BMC Series Editorial Production Team. We have responded to each of the suggestions and have modified the manuscript accordingly so that the present version should satisfy their requirements. Namely,

1. We removed Figure 5 which was mentioned in between Figures 1 and 2 on page 6 of the manuscript and added an item 1f to the Figure 1 and corrected according to the change the text on page 6 and Figure legend for Figure 1.
2. Figure 6 was renamed Figure 5.
3. We removed additional files.

The authors thank you and the reviewers for your time and commitment.

Sincerely,

Kosaku Iwatsubo, M.D., Ph.D.
Corresponding author